Author's response to reviews

Title: Prognostic Factors and Multidisciplinary Treatment Modalities of Brain Metastases from Colorectal Cancer: Analysis of 93 Patients

Authors:

Xiao-Dong Gu (061105315@fudan.edu.cn)
Yan-Tao Cai (yantaocai@hotmail.com)
Yi-Ming Zhou (hankunh@sohu.com)
Zhen-Yang Li (gxdgxd737@sohu.com)
Jian-Bin Xiang (68875417@qq.com)
Zong-You Chen (zongyouc@sohu.com)

Version: 2
Date: 2 July 2014

Author's response to reviews:

1. Line numbering is provided.
2. A statement is added in the Methods section of the manuscript.